Author’s response to reviews

Title: Risk factors and psychosocial characteristics of problematic and potential problematic internet use among adolescents: A cross-sectional study

Authors:

George Kormas (gkormas@gmail.com)
Elena Critselis (ecritselis@yahoo.com)
Mari Janikian (mjanikian@acq.edu)
Dimitrios Kafetzis (kafetzis@ath.forthnet.gr)
Artemis Tsitsika (info@youth-health.gr)

Version: 4 Date: 19 July 2011

Author’s response to reviews: see over
Dear Prof Sigfusdottir and BMC Team,

Thank you for accepting our manuscript entitled “Risk factors and psychosocial characteristics of potential problematic and problematic internet use among adolescents: A cross-sectional study” for publication in the Journal of BMC Public Health. Please find below a detailed point-by-point response (in bullets and italics) to all formatting changes incorporated in the revised manuscript.

Sincerely,

Artemis Tsitsika
Lecturer in Pediatrics-Adolescent Medicine
Adolescent Health Unit (AHU)
Second University Department of Pediatrics, “P. & A. Kyriakou” Children’s Hospital, National and Kapodistrian University of Athens School of Medicine, Athens, Greece
Formatting Changes

Major revisions (we require the author to make these changes)

Title: please ensure that the title is identical in the manuscript and the submission system
Thank you for noting this inconsistency. Please find that this correction has been made in the title so that it is identical in both the manuscript and the submission system.

Tables: Please remove any colours from the tables. We cannot display these.
Please find that this correction has been made and the colours have been removed from the tables.

Authors' contributions: Please indicate that all authors read and approved the final manuscript.
Please find that a sentence was included in the Authors’ Contributions section that “All authors read and approved the final manuscript.”

Tables: Please remove all visible vertical lines from tables.
Please find that this correction has been made and all visible vertical lines have been removed from the tables.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Highlighting/tracking: Please remove all highlighting/tracking from the manuscript.
Please find that this correction has been made and all highlighting/tracking parts have been cleared from the manuscript.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
Please note that we have checked our manuscript for typographical errors.